%% start of file `template.tex'.
%% Copyright 2006-2013 Xavier Danaux (xdanaux@gmail.com).
%
% This work may be distributed and/or modified under the
% conditions of the LaTeX Project Public License version 1.3c,
% available at http://www.latex-project.org/lppl/.


\documentclass[11pt,a4paper,sans]{moderncv}        % possible options include font size ('10pt', '11pt' and '12pt'), paper size ('a4paper', 'letterpaper', 'a5paper', 'legalpaper', 'executivepaper' and 'landscape') and font family ('sans' and 'roman')

% moderncv themes
\moderncvstyle{classic}                             % style options are 'casual' (default), 'classic', 'oldstyle' and 'banking'
\moderncvcolor{blue}                               % color options 'blue' (default), 'orange', 'green', 'red', 'purple', 'grey' and 'black'
%\renewcommand{\familydefault}{\sfdefault}         % to set the default font; use '\sfdefault' for the default sans serif font, '\rmdefault' for the default roman one, or any tex font name
%\nopagenumbers{}                                  % uncomment to suppress automatic page numbering for CVs longer than one page

% character encoding
\usepackage[utf8]{inputenc}                       % if you are not using xelatex ou lualatex, replace by the encoding you are using
\usepackage[ngerman]{babel}
%\usepackage{CJKutf8}                              % if you need to use CJK to typeset your resume in Chinese, Japanese or Korean

% adjust the page margins
\usepackage[scale=0.75]{geometry}
%\setlength{\hintscolumnwidth}{3cm}                % if you want to change the width of the column with the dates
%\setlength{\makecvtitlenamewidth}{10cm}           % for the 'classic' style, if you want to force the width allocated to your name and avoid line breaks. be careful though, the length is normally calculated to avoid any overlap with your personal info; use this at your own typographical risks...

% personal data
\name{Fynn}{Fabry}
\title{CV}                               % optional, remove / comment the line if not wanted
\address{Okerstr. 15}{12049 Berlin}{}% optional, remove / comment the line if not wanted; the "postcode city" and and "country" arguments can be omitted or provided empty
\phone[mobile]{+49~(177)~270~189~8}                   % optional, remove / comment the line if not wanted
\email{fynn@someserver.de}                               % optional, remove / comment the line if not wanted
%\homepage{www.johndoe.com}                         % optional, remove / comment the line if not wanted
%\extrainfo{additional information}                 % optional, remove / comment the line if not wanted
%\photo[64pt][0.4pt]{headshot.jpg}                       % optional, remove / comment the line if not wanted; '64pt' is the height the picture must be resized to, 0.4pt is the thickness of the frame around it (put it to 0pt for no frame) and 'picture' is the name of the picture file
%\quote{Some quote}                                 % optional, remove / comment the line if not wanted

% to show numerical labels in the bibliography (default is to show no labels); only useful if you make citations in your resume
%\makeatletter
%\renewcommand*{\bibliographyitemlabel}{\@biblabel{\arabic{enumiv}}}
%\makeatother
%\renewcommand*{\bibliographyitemlabel}{[\arabic{enumiv}]}% CONSIDER REPLACING THE ABOVE BY THIS

% bibliography with mutiple entries
%\usepackage{multibib}
%\newcites{book,misc}{{Books},{Others}}
%----20.11.2018--------------------------------------------------------------------
%            content
%----------------------------------------------------------------------------------
\begin{document}
%\begin{CJK*}{UTF8}{gbsn}                          % to typeset your resume in Chinese using CJK
%-----       letter       ---------------------------------------------------------
% recipient data
\recipient{Christine Schiesser}{Ullsteinstraße\\Ullsteinstraße 130\\Berlin 12109}
\date{20.01.2020}
\opening{Liebe Christine,}
\closing{Viele Grüße}
\enclosure[Anlagen]{Lebenslauf Deutsch, Lebenslauf Englisch}          % use an optional argument to use a string other than "Enclosure", or redefine \enclname
\makelettertitle

Über Freunde wurde ich vor Kurzem Daniel vorgestellt, der mir von der Arbeit bei SecConsult erzählte. Er klang dabei sehr begeistert von der Arbeit bei SecConsult und was er mir beschrieben hat, klang für mich genau nach dem beruflichen Umfeld, nach dem ich Suche. Als Security Consultant schätze ich die Nähe zum Kunden, die sowohl bei Daniel als auch in meinem Gespräch mit Julian immer wieder betont wurden. Weiterhin überzeugt hat mich die Aussicht auf aktive Beteiligung und später eigenständige Durchführung von Pentests.

Aktuell arbeite ich als Security Consultant und führe vornehmlich Tests durch, die die Netzwerüberwachung von Kundenunternehmen testen. Aufbauend auf meiner Erfahrung in der Entwicklung von Angriffsvektoren und 

\makeletterclosing

\clearpage
%-----       resume       ---------------------------------------------------------
\makecvtitle

\section{Berufserfahrung}
\cventry{Seit Jan 2019}{Security Consultant}{Security Research Labs}{Berlin}{}{Entwicklung eines Tools }

\section{Ausbildung}
\cventry{2009--2018}{1. Staatsexamen Mathematik/Physik Lehramt für Gmynasium/Gesamtschule}{RWTH Aachen}{Aachen}{}{}
%\cventry{year--year}{Degree}{Institution}{City}{\textit{Grade}}{Description}

%\section{Thesis}
%\cvitem{}{\emph{Title}}
%\cvitem{supervisors}{Supervisors}
%\cvitem{description}{Short thesis abstract}

\section{Außercurriculare Tätigkeiten}
%\cventry{seit 2009}{Student (Staatsexamen)}{RWTH Aachen University}{Aachen}{}{Studium Mathematik/Physik auf Lehramt\newline{} Schulform Gymnasium/Gesamtschule}
\subsection{Arbeit als studentische Hilfskraft}
\cventry{2010--2011}{Förderlehrkraft}{Lerntherapeutisches Institut für Leserechtschreibschwäche und Dyskalkulie}{Aachen}{}{Förderunterricht für Kinder mit Dyskalkulie oder Leserechtschreibschwäche}%
%Detailed achievements:%
%\begin{itemize}%
%\item Achievement 1;
%\item Achievement 2, with sub-achievements:
%  \begin{itemize}%
%  \item Sub-achievement (a);
%  \item Sub-achievement (b), with sub-sub-achievements (don't do this!);
%    \begin{itemize}
%    \item Sub-sub-achievement i;
%    \item Sub-sub-achievement ii;
%    \item Sub-sub-achievement iii;
%    \end{itemize}
%  \item Sub-achievement (c);
%  \end{itemize}
%\item Achievement 3.
%\end{itemize}}
\cventry{2013--2014}{Praktikumsbetreuerin}{RWTH Aachen}{Aachen}{}{Vorbereitung und Beaufsichtigung von Praktikumsversuchen für Medizin- und Biologiestudierende}
\cventry{2014--2015}{Programmiererin}{RWTH Aachen}{Aachen}{}{Implementierung von Optimierungsalgorithmen und Simulation von Mobilfunknetzwerken in C++ mit Gurobi}
\subsection{Ehrenamtliche und sonstige Tätigkeiten}
%\cventry{2009--2011}{Gruppenleiterin, Kanzlerin}{KSJ}{Geilenkirchen}{}{Organisation und Leitung von Sommerlagern, Wochenendfreizeiten und Gruppenstunden, Buchhaltung}
\cventry{2012--2015}{Fachschaftsarbeit}{Fachschft I/1 der RWTH Aachen}{Aachen}{}{Organisation und Durchführung einer Studierendenkonferenz 2015, studentische Selbstorganisation, Beratung zu Studienthemen}
\cventry{2013--2017}{Technischer/Organisatorischer Helfer}{Linux Install Parties}{Aachen}{}{Support bei Erstinstallation und Verwendung von Ubuntu Linux im Dualboot für Studierende}
\cventry{2017}{Zeichnerin, Setzerin}{Selbstverlag}{Aachen}{}{Grafische Umsetzung, Satz, Produktion und Vertrieb eines Comics}
\cventry{seit 2017}{Participant}{Wargames}{}{}{Themenbereiche: Binary Exploitation, Kryptoanalyse, Websicherheit, Embedded Security}

\section{Computerbasierte Fähigkeiten}
\cvitem{Anwendung}{Datenauswertung und -darstellung (Excel), UNIX/Linux, Git, Powerpoint, diverse shellbasierte Anwendungen}
\cvitem{Programmieren}{Häufige Verwendung von Python und BASH, Grundkenntnisse in Ruby, C++, PHP und SQL}
\cvitem{Textsatz}{Häufige bis tägliche Verwendung von \LaTeX{} und Vim}
%\cvitem{IT Support}{Grundlagensupport in Ubuntu Linux und gängigen Anwendungen}
\cvitem{Grafikdesign}{Gestaltung von Postern, Flyern und Comics (Inkscape, GIMP, Krita)}

\clearpage
\section{Sprachen}
\cvitemwithcomment{Deutsch}{Muttersprache}{}
\cvitemwithcomment{Englisch}{Flüssig}{}
\cvitemwithcomment{Französisch}{Grundkenntnisse}{}

\section{Interessen}
\cvitem{}{Zeichnen/Malen (Portraits, Szenerie, Comics)}
\cvitem{}{Lockpicking}
\cvitem{}{Klettern}
%\cvitem{Bouldern}{Meistens in der Halle, manchmal auch draußen. Gelegentlich auch Toprope-Klettern.}

%\section{Thesis}
%\cvlistitem{Item 1}
%\cvlistitem{Item 2}
%\cvlistitem{Item 3. This item is particularly long and therefore normally spans over several lines. Did you notice the indentation when the line wraps?}
%
%\section{Extra 2}
%\cvlistdoubleitem{Item 1}{Item 4}
%\cvlistdoubleitem{Item 2}{Item 5\cite{book1}}
%\cvlistdoubleitem{Item 3}{Item 6. Like item 3 in the single column list before, this item is particularly long to wrap over several lines.}

%\section{References}
%\begin{cvcolumns}
%  \cvcolumn{Category 1}{\begin{itemize}\item Person 1\item Person 2\item Person 3\end{itemize}}
%  \cvcolumn{Category 2}{Amongst others:\begin{itemize}\item Person 1, and\item Person 2\end{itemize}(more upon request)}
%  \cvcolumn[0.5]{All the rest \& some more}{\textit{That} person, and \textbf{those} also (all available upon request).}
%\end{cvcolumns}

% Publications from a BibTeX file without multibib
%  for numerical labels: \renewcommand{\bibliographyitemlabel}{\@biblabel{\arabic{enumiv}}}% CONSIDER MERGING WITH PREAMBLE PART
%  to redefine the heading string ("Publications"): \renewcommand{\refname}{Articles}
\nocite{*}
\bibliographystyle{plain}
\bibliography{publications}                        % 'publications' is the name of a BibTeX file

% Publications from a BibTeX file using the multibib package
%\section{Publications}
%\nocitebook{book1,book2}
%\bibliographystylebook{plain}
%\bibliographybook{publications}                   % 'publications' is the name of a BibTeX file
%\nocitemisc{misc1,misc2,misc3}
%\bibliographystylemisc{plain}
%\bibliographymisc{publications}                   % 'publications' is the name of a BibTeX file

\clearpage
%\clearpage\end{CJK*}                              % if you are typesetting your resume in Chinese using CJK; the \clearpage is required for fancyhdr to work correctly with CJK, though it kills the page numbering by making \lastpage undefined
\makecvtitle

\section{Carreer}
\cventry{Since Jan 2019}{Security Consultant}{Security Research Lab}{Berlin}{}{}

\section{Education}
\cventry{2009--2018}{1. State Examination in Mathematics/Physics Teacher's Studies}{RWTH Aachen}{Aachen}{}{}
%\cventry{year--year}{Degree}{Institution}{City}{\textit{Grade}}{Description}

%\section{Thesis}
%\cvitem{}{\emph{Title}}
%\cvitem{supervisors}{Supervisors}
%\cvitem{description}{Short thesis abstract}

\section{Non-curricular Activities}
%\cventry{seit 2009}{Student (Staatsexamen)}{RWTH Aachen University}{Aachen}{}{Studium Mathematik/Physik auf Lehramt\newline{} Schulform Gymnasium/Gesamtschule}
\subsection{Employment as Student Assistant}
\cventry{2010--2011}{Support Teacher}{Lerntherapeutisches Institut für Leserechtschreibschwäche und Dyskalkulie}{Aachen}{}{Additional lessons for children with dyscalculia or dyslexia}%
%Detailed achievements:%
%\begin{itemize}%
%\item Achievement 1;
%\item Achievement 2, with sub-achievements:
%  \begin{itemize}%
%  \item Sub-achievement (a);
%  \item Sub-achievement (b), with sub-sub-achievements (don't do this!);
%    \begin{itemize}
%    \item Sub-sub-achievement i;
%    \item Sub-sub-achievement ii;
%    \item Sub-sub-achievement iii;
%    \end{itemize}
%  \item Sub-achievement (c);
%  \end{itemize}
%\item Achievement 3.
%\end{itemize}}
\cventry{2013--2014}{Lab Teaching Assistant}{RWTH Aachen}{Aachen}{}{Preparation, supervision and evaluation of physics lab courses for biology students and medical students}
\cventry{2014--2015}{Programmer}{RWTH Aachen}{Aachen}{}{Implementation of algorithms for network optimization problems, working with C++, Lemon and Gurobi}
\subsection{Volunteering and Other Activities}
%\cventry{2009--2011}{Gruppenleiterin, Kanzlerin}{KSJ}{Geilenkirchen}{}{Organisation und Leitung von Sommerlagern, Wochenendfreizeiten und Gruppenstunden, Buchhaltung}
\cventry{2012--2015}{Students' Association Member}{Fachschft I/1 der RWTH Aachen}{Aachen}{}{Organisation of a student conference in 2015, study guidance and student self-administration}
\cventry{2013--2017}{Technical/Organisational Support}{Linux Install Parties}{Aachen}{}{Support for students during first time Linux installation and introduction to various Ubuntu-supported programs}
\cventry{2017}{Artist, Typesetter}{Selfpublishing}{Aachen}{}{Planning, graphic realization and marketing of a comic project.}
\cventry{since 2017}{Participant}{Wargames}{}{}{Topics: Binary Exploitation, cryptanalysis, websecurity, embedded security}

\section{Computer Skills}
\cvitem{Applications}{Various commandline applications, Git, UNIX/Linux, data evaluation and visualization (Excel)}
\cvitem{Programming}{Frequent use of Python and BASH, basic knowledge of Ruby, C++, Java, PHP and SQL}
\cvitem{Editing}{Daily use of \LaTeX{} and Vim}
%\cvitem{IT Support}{Grundlagensupport in Ubuntu Linux und gängigen Anwendungen}
\cvitem{Grafic design}{Design of posters, flyers and comics (Inkscape, GIMP, Krita)}

\clearpage
\section{Languages}
\cvitemwithcomment{German}{Native}{}
\section{Further Interests}
\cvitem{}{Drawing/Painting (Portraits, Scenery, Comics)}
\cvitem{}{Lockpicking}
\cvitem{}{Climbing}
%\cvitem{Bouldern}{Meistens in der Halle, manchmal auch draußen. Gelegentlich auch Toprope-Klettern.}

%\section{Thesis}
%\cvlistitem{Item 1}
%\cvlistitem{Item 2}
%\cvlistitem{Item 3. This item is particularly long and therefore normally spans over several lines. Did you notice the indentation when the line wraps?}
%
%\section{Extra 2}
%\cvlistdoubleitem{Item 1}{Item 4}
%\cvlistdoubleitem{Item 2}{Item 5\cite{book1}}
%\cvlistdoubleitem{Item 3}{Item 6. Like item 3 in the single column list before, this item is particularly long to wrap over several lines.}

%\section{References}
%\begin{cvcolumns}
%  \cvcolumn{Category 1}{\begin{itemize}\item Person 1\item Person 2\item Person 3\end{itemize}}
%  \cvcolumn{Category 2}{Amongst others:\begin{itemize}\item Person 1, and\item Person 2\end{itemize}(more upon request)}
%  \cvcolumn[0.5]{All the rest \& some more}{\textit{That} person, and \textbf{those} also (all available upon request).}
%\end{cvcolumns}

% Publications from a BibTeX file without multibib
%  for numerical labels: \renewcommand{\bibliographyitemlabel}{\@biblabel{\arabic{enumiv}}}% CONSIDER MERGING WITH PREAMBLE PART
%  to redefine the heading string ("Publications"): \renewcommand{\refname}{Articles}
\nocite{*}
\bibliographystyle{plain}
\bibliography{publications}                        % 'publications' is the name of a BibTeX file

% Publications from a BibTeX file using the multibib package
%\section{Publications}
%\nocitebook{book1,book2}
%\bibliographystylebook{plain}
%\bibliographybook{publications}                   % 'publications' is the name of a BibTeX file
%\nocitemisc{misc1,misc2,misc3}
%\bibliographystylemisc{plain}
%\bibliographymisc{publications}                   % 'publications' is the name of a BibTeX file

\clearpage

\end{document}


%% end of file `template.tex'.
